# RESEARCH

**Open Access** 

# Attitudes and gaps in knowledge of the diagnosis, treatment, and psychopathology of eating disorders among different health professionals

Check for updates

Alessio Maria Monteleone<sup>1\*</sup>, Marco Carfagno<sup>1</sup>, Eugenia Barone<sup>1</sup>, Giammarco Cascino<sup>2</sup>, Armando Pitocco<sup>1</sup>, Carlotta Brandi<sup>1</sup>, Lorenzo Landolfi<sup>1</sup>, Claudia Toni<sup>1</sup>, Gaia Sampogna<sup>1</sup> and Andrea Fiorillo<sup>1</sup>

# Abstract

**Background** Health professionals from different specialties in medical and psychological areas play an important role in diagnosis and treatment of eating disorders (EDs). This study aimed to identify gaps in knowledge about the diagnosis, etiology, and management of EDs and to assess health professionals' attitudes towards these illnesses.

**Methods** A new questionnaire was developed and validated. Residents and consultants working in disciplines involved in the management of EDs (namely, internal medicine, general practitioners, psychiatric area, psychological area, and surgical area) completed the questionnaire. Knowledge and attitudes were compared among the study groups through one-way ANCOVA and chi-square tests.

**Results** The final version of the questionnaire consisted of 54 items assessing the following areas: stigma, treatment, physical complications, diagnosis, and aetiopathogenesis of EDs. For all health professionals the area of most deficiency was the aetiopathogenesis, while the best one was the management of physical complications. All medical professionals showed less knowledge than psychiatrists in terms of etiology, diagnosis, and treatment of EDs. A lack of knowledge about evidence-based psychotherapies, general psychopathology, and family members' role in the management of EDs emerged among all health professionals. Stigma was found among non-mental health professionals who considered these patients to be different from others and responsible for their abnormal eating behaviors.

**Conclusions** Clarifying the health professionals' specific gaps occurring in the knowledge of EDs and in the attitudes towards these individuals may inform educational programs to improve early detection and management of EDs.

# **Plain English summary**

This article investigated attitudes towards individuals with eating disorders (EDs) and knowledge of the diagnosis and management of these illnesses among health professionals. A gap in the knowledge of diagnosis, aetiopathogenesis, and treatment emerged above all in non-mental health professionals. Psychotherapy

\*Correspondence: Alessio Maria Monteleone alessiomaria.monteleone@unicampania.it

Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicate dot events in a credit line to the material. If material is not included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

effectiveness and the role of family members in the therapeutic process were not sufficiently acknowledged, and general psychological factors contributing to the onset of EDs were not recognized. Impaired attitudes were primarily observed among surgeons, although all health professionals considered these disorders as distinct from others and viewed these individuals as responsible for their abnormal eating behaviors. These findings outline a type of stigma towards EDs that is associated with health professionals and may impair early diagnosis and recovery. Educational programs should aim to provide continuous education to update and improve the knowledge of EDs among health professionals.

Keywords Eating disorders, Anorexia nervosa, Stigma, Education, Health professional

# Background

Eating disorders (EDs) are complex psychiatric disorders with high rate of psychiatric comorbidities, physical complications, and social impairment [1]. Although the highly prevalent among adolescents and young adults, EDs can occur across all ages, as well as ethnicities and social levels [2]. Despite a decreasing mortality rate over time [3], AN still has the highest rate of mortality among psychiatric disorders [4], with a high rate of chronicity and inadequate recovery [3]. Children and adolescents have better outcomes and a lower rate of chronicity than adults across and within EDs [3]. Illness duration is a negative prognostic factor [5], and a staging model of the illness has been suggested [6]: while ED progresses, several psychosocial and neurobiological factors make the persons resistant to treatment, promoting the persistence of the illness [6]. This supports the need for patients with EDs to be identified early and moved to specialist care as quickly as possible. The guidelines suggest that treatment should be provided at the first detection [7].

Despite this recommendation, only 20-25% of individuals access specialist care for their symptoms [8]. This problem is favored by the complexity of EDs, which require a multidisciplinary approach to address physical, behavioral, and psychosocial problems [9]. Further barriers include lack of awareness of the illness, mental health-related stigma, family knowledge about the disorder, fragmented youth mental health care and organization of the pathways that lead individuals to seek care for their ED [8, 10]. A recent European multicenter study showed that the involvement of many health professionals and the occurrence of affective symptoms (i.e., anxious and depressive symptoms) delay access to specialist ED units [11]. In addition to mental health professionals, general practitioners play a central role in early diagnosis and treatment promotion for EDs [11]. Individuals with EDs consult general practitioners more frequently than those without EDs in the five years before receiving the ED diagnosis [12] and have a high rate of access to mental and physical health services [13]. Involving non-mental health professionals is important both for primary care diagnosis and for treatment: a higher number of specialist treatment components are associated with higher rates of recovery, and a multidisciplinary treatment approach predicts better outcomes in EDs [3].

This claims that an accurate knowledge of ED assessment and treatment is essential among health professionals. Unfortunately, clinicians who are not not specialized in EDs did not show better knowledge about EDs than students or individuals with EDs [14]. Clinicians claim that they lack the necessary skills to intervene when treating individuals with EDs [15] and report stress and negative reactions such as hopelessness, lack of competence, and worry that are associated with their stigmatizing beliefs, inexperience, and gender [16, 17]. Their lack of confidence in helping individuals with EDs may be due to their lack of necessary training [18]: this may result in treatment delays or inappropriate management. Psychiatrists also felt confident in diagnosing EDs, while they reported low confidence in the management of EDs [19].

Given this background, it is important to investigate the knowledge gap between health professionals, considering the differences among clinical specialists. By addressing this gap, educational training could be improved. The primary objective of this study was to identify specific areas where health professionals have inadequate knowledge about EDs, including stigma beliefs, assessment procedures, managing physical complications, treatment options, and risk factors. The second objective of the study was to examine whether there are differences in knowledge and attitudes regarding EDs among health professionals involved in ED management (namely, mental health, primary care, internal medicine, surgical area).

# Methods

## Measures and procedure

The study included three steps. First, a panel composed of four researchers and clinicians of varying seniority ranging from residents to consultants and working in the ED field generated the items of the questionnaire. The items were developed from the *Questionario sulle Opinioni degli Italiani riguardo a.i. disturbi mentali* [20], which was validated for the assessment of clinicians' knowledge and stigma towards serious mental illnesses. The items were modified considering the Italian guidelines for diagnosis and management of EDs [21], the DSM-5 ED diagnostic criteria [22] and research team working experience. The first version of the questionnaire consisted of 73 items. In the second stage, the original items were reviewed by three expert psychiatrists and were critically discussed, reviewed and, when necessary, rewritten to improve clarity and readability.

The next steps involved recruiting people who were given the pre-final version of the questionnaire, the analysis of the main psychometric properties of the instrument, and the finalization of the questionnaire.

The study was open to participants who worked at University of Campania L. Vanvitelli and met the following inclusion criteria: (a) working as a resident or a consultant in one of the following disciplines: psychiatry, child neuropsychiatry, pediatrics, cardiology, dermatology, endocrinology, gastroenterology, internal medicine, nephrology, neurology, general surgery, gynecology, orthopedics, urology; b) being a psychologist; and c) being a general practitioner working as supervisor for medical students. A convenience sampling method was used to enroll participants, who received an email that included a link to an online questionnaire. The minimum number of respondents was 365, given that validating a questionnaire requires 5 respondents for each item [23].

Participants were more frequently female (58%), with a mean age of 34.2 ( $\pm$ 10.2) yrs. 12% were specialized in a surgical discipline (i.e., general surgeons, gynecologists, orthopedics, and urologists), 32.6% were included in the general practitioner area (i.e., general practitioners and pediatricians), 29% were specialized in a discipline belonging to the internal medicine area (i.e., cardiologists, dermatologists, endocrinologists, gastroenterologists, internists, nephrologists, and neurologists), 17% were included in the psychiatric area (i.e., psychiatrists and child neuropsychiatrists) and 8% were included in the psychological area (i.e., psychologists).

Participants filled in the preliminary version of the questionnaire and provided a rating of importance and appropriateness for each item on a 10-level scale (1=item not important at all; 10=item very important). They were asked to provide dichotomous (i.e., yes, no), categorical (i.e., yes always, yes sometimes, rarely, never), or multiple answers. A full (namely, 1 point) score was assigned to correct answers. Twenty-five participants were asked to assess the test-retest reliability. The reasons for discrepancies in the test-retest group were explored through an ad hoc schedule and discussed among researchers. Ethics approval was not required for this kind of survey as per local legislation and national guidelines.

# Statistical analysis

The face validity of the items was explored by means of the ratings on the 1-10 appropriateness scale. Pearson's rho coefficient was used to evaluate the variance of the

responses. The test-retest reliability of each item was analyzed through Cohen's kappa coefficient. Cronbach's alpha analysis was used to group single items into the hypothesized subscales (content validity).

The rate of correct answers in the entire sample was calculated. A one-way ANCOVA followed by a post hoc Tukey's test was conducted to compare the general score and those of each assessed topic (i.e., A, B, C, D, E) among the study groups (i.e., surgical area, internal medicine, general practitioners, psychiatric area and psychological area) controlling for age. The topic scores were calculated as the sum of each item included in that topic area.

The chi-square test was performed to compare the rate of correct answers among the study groups. To explore the differences in terms of stigma and deficit of knowledge, the chi-square test was performed only for items with a frequency of correct answers below 67% in the entire sample. A post hoc test with Bonferroni corrections was performed to examine the differences between study groups. The level of significance was set at p < .05. All analyses were performed using JASP software [24].

# Results

The final study sample consisted of 405 participants. Participants compiled the prefinal version of the questionnaire consisting of 73 items. From this list, 7 items rated as not important or relevant (i.e., with an appropriateness rating < 6.0) were eliminated, and 12 items were deleted due to low reliability (7 items with Cohen's kappa coefficient < 0.60; 5 items with Pearson's rho coefficient < 0.7). The final version of the questionnaire consisted of 54 items including dichotomous and multiple-choice questions scoring from 0 (totally wrong) and 1 (totally correct), which could be grouped into five categories based on the topic, i.e., stigma (A score), treatment (B score), physical complications (C score), diagnosis (D score), aetiopathogenesis (E score), and general knowledge (Total score). The items referred to an example of ED.

The results of the one-way ANCOVA are reported in Table 1.

When the A score (Cronbach's  $\alpha$ =0.48) was entered in the model as dependent variable, the post hoc analysis showed a significant difference in terms of stigma between the surgical area and psychiatric area (*p*=.007), with psychiatrists reporting lower levels of stigma. Regarding the B score (Cronbach's  $\alpha$ =0.42), the post hoc analysis indicated a significant difference in the knowledge of treatment of EDs, with a better knowledge found in the psychiatric area compared to surgical (*p*=.003), internal medicine (*p*<.001) and general practitioner (*p*<.001) areas. Psychologists also showed better knowledge of treatment compared to surgical (*p*=.006), internal medicine (*p*=.003) and general practitioner areas

 Table 1
 Comparison of assessed topics among health professional areas

Торіс	Gen (mean±SD)	Med (mean±SD)	Psyl (mean±SD)	Psyt (mean±SD)	Surg (mean±SD)	ANCOVA
Stigma	$7.583 \pm 1.683$	$7.534 \pm 1.451$	$7.343 \pm 1.168$	$7.831 \pm 1.483$	6.724±2.199	Age F(1,399) = 11.756 p < .001
						Study groups F(4,399) = 3.008 p = .018
Treatment	$7.368 \pm 1.266$	$7.327 \pm 1.112$	$8.064 \pm 1.037$	$8.030 \pm 1.032$	$7.174 \pm 1.407$	Age F(1,399) = 10.14 p = .002
						Study groups F(4,399) = 8.065 <i>p</i> < .001
Complications	$7.153 \pm 1.147$	$7.316 \pm 0.987$	$5.557 \pm 1.498$	$7.817 \pm 0.950$	$7.296 \pm 1.040$	Age F(1,399) = 1.659 p = .198
						Study groups F(4,399) = 25.167 <i>p</i> < .001
Diagnosis	$7.89 \pm 1.429$	$7.898 \pm 1.347$	$7.486 \pm 1.353$	$8.761 \pm 1.281$	$7.429 \pm 1.851$	Age F(1,399) = 44.055 p < .001
						Study groups F(4,399) = 7.825 <i>p</i> < .001
Aetiopathogenesis	$7.848 \pm 1.884$	$8.051 \pm 1.894$	8.314±1.952	$8.915 \pm 2.103$	$7.714 \pm 2.062$	Age F(1,399) = 4.509 p = .034
						Study groups F(4,399) = 4.295 <i>p</i> = .002
Total	$37.84 \pm 4.598$	$38.125 \pm 4.128$	$36.764 \pm 3.939$	$41.354 \pm 4.826$	$36.338 \pm 6.257$	Age F(1,399) = 27.414 p < .001
						Study groups F(4,399) = 10.531 <i>p</i> < .001

Gen=general practitioner area; Med=internal medicine area; Psyl=psychological area; Psyt=psychiatric area; Surg=surgical area

(p=.005). When the C score (Cronbach's  $\alpha$ =0.36) was entered in the model as dependent variable, the post hoc analysis showed a significant difference in terms of knowledge of physical complications between psychiatric and all other areas, with psychiatrists reporting a better understanding and management of complications compared to general practitioner (p < .001), internal medicine (p=.022), and psychological (p<.001) areas. In contrast, psychologists showed lower levels of expertise on complications compared to surgical (p < .001), general practitioner (p<.001), and internal medicine (p<.001) areas. When post hoc analysis was conducted for the D score (Cronbach's  $\alpha = 0.47$ ), a significant difference in the level of expertise in the diagnosis of EDs has emerged, with psychiatrists showing better knowledge than health professionals included in surgical (p < .001), general practitioner (p < .001), internal medicine (p < .001), and psychological (p=.001) areas. The post hoc analysis computed for the E score (Cronbach's  $\alpha = 0.5$ ) revealed a significant difference in the level of expertise in the aetiopathogenesis of EDs, with psychiatric area showing better knowledge compared to surgical (p=.02), internal medicine (p=.029), and general practitioner (p=.002) areas. When the total score (Cronbach's  $\alpha = 0.71$ ) was entered in the model as dependent variable, the post hoc analysis revealed a significant difference in terms of general knowledge of EDs between psychiatric and all other areas, with psychiatrics showing the best results (p < .001).

The rate of correct answers for each item in the entire sample is reported in Table 2.

Rates of errors were compared among health professional areas through chi-square tests: the results are shown in Table 3. Significant results were found for the following items: psychotherapy intervention could help people with EDs (item 9); there are psychotherapy interventions specifically validated for EDs (item 10); nutritional intervention is the first step to help people with EDs (item 11); instrumental examinations useful to evaluate physical complications (item 17); amenorrhea is a mandatory diagnostic criterion (item 22); it is easy to perceive if a person has ever had an ED (item 28); an individual with a family member suffering from EDs is more likely to develop the disorder (item 34); EDs can be due to pathological perfectionism (item 46); EDs can be due to drug abuse (item 48); and EDs can be due to friendship with wrong or untrustworthy people (item 49).

# Discussion

Knowledge and stigma towards EDs among health professionals were assessed in this study. All health professionals experienced difficulties with ED diagnosis and management compared to psychiatrists, while only surgeons displayed heightened stigma compared to psychiatrists. While physical complications management was the area of greatest expertise for health professionals, knowledge of risk factors was the area with the greatest deficiency. Better knowledge and less stigma were linked to higher participant age. In each study group there were specific areas of impairment in diagnosing and treating EDs, but all health professionals failed to recognize the role of family members in managing EDs.

The diagnosis of EDs was problematic for all health professionals, compared to psychiatrists. Early ED diagnosis is crucial to provide early care and promote better outcomes [3]. Individuals with EDs often require help for their somatic problems and access to physical health services more than to psychiatric services in the year preceding their ED diagnosis [25]. Thus, general practitioners, pediatricians and hospital doctors play an important role in diagnosis, and their difficulties with diagnosis can delay access to specialized care. This is supported by a multicenter European study showing that the cooccurrence of somatic symptoms at clinical referral is

**Table 2** Topics of the questionnaire with frequency of right answers for each item in the all sample. The items refer to the description of the following clinical case.In certain periods of their lives, some individuals think or say to be overweight, are dissatisfied with their bodies, restrict food intake, and overthink about food or deny the evidence that they are losing weight. Their behaviours sometimes differ from those observed in most people: they prefer eating alone or, when they think to have failed with their eating patterns and rules or to have lost control on food intake, can self-induce vomiting, abuse laxatives, or do excessive and compulsive physical exercise

ITEM		Frequen- cy of right answers (%)
Stigma	2	
2	May a person with symptoms described above suffer from?	99.5
26	Are diseases such as those described above like any other disease (e.g., diabetes, heart disease, etc.)?	54.2
27	Are diseases such as those described above like any other mental disorder (e.g., depression, schizophrenia, anxiety)?	67.7
28	Is it easy to notice if a person has ever had disorders such as those described above?	46.1
29	Do successful people rarely suffer from disorders such as those described above?	84.7
31	Should people with disorders such as those described above have not children?	91.4
33	Should the National Health System spend more money for the care of people with problems such as those described above?	69
40	Is it possible to diagnose a disorder such as those described above through the person's appearance?	60.8
41	Do people with disorders such as those described above suffer from a mental disorder?	79.3
42	Have people with disorders such as those described above chosen to behave in that way?	64.3
Treatm	nent	
4–5	Do you think that psychotropic drugs are effective in the treatment of disorders such as those described above? If not, which is the most effective treatment for these disorders?	94.1
6	What treatments should be prescribed to people with disorders such as those described above?	98.3
7	Do you think a multidisciplinary team approach could be useful for people with disorders such as those described above?	89.7
8	If so, which health professionals should be included in that team?	95.1
9	Do you think psychotherapy could help people with disorders such as those described above?	60.1
10	Do you think there are specific evidence-based psychotherapies for disorders such as those described above?	34.5
11	Do you think that a nutritional intervention is the first step to help people with disorders such as those described above?	40.4
12	In which hospital setting patients with disorders such as those described above should be treated?	88.7
32	Should family members be included in the treatment of people with disorders such as those described above?	49.5
39	May patients suffering from a disorder such as those described above and comorbid diabetes be treated with insulin?	13.1
54	Which are the effects of psychotherapy in disorders such as those described above?	70.9
Physica	al complications	
13	May people with disorders such as those described above develop not severe medical complications, i.e., not life-threatening?	76.1
14	In the presence of underweight, which are the most frequent physical complications?	68.2
15	Would you prescribe blood tests to a person with disorders such as those described above to diagnose medical complications?	86
16	Which of these blood tests would you prescribe for a person with disorders such as those described above?	95.8
17	Which of these medical tests could be useful for a person with disorders such as those described above?	64.5
19	Is it important to measure serum electrolytes in people with disorders such as those described above?	89.2
20	If a person with disorders such as those described above has leucopenia, this is most frequently an indication of $\ldots$	78.8
21	If a person with disorders such as those described above and underweight has heart rate at 52 beats per minute, it is most frequently an indication of	74.6
23	ls estroprogestinic replacement therapy always indicated in patients with disorders such as those described above and amenorrhea?	76.8
Diagno	osis	
1	May a person with the symptoms of the disorder described above suffer from one of the following disorders?	95.3
3	If yes, which health professional should treat a person with the symptoms of the disorder described above?	86.9
18	May normal blood tests allow us to rule out a disorder such as those described above?	42.4
22	Is amenorrhea a mandatory diagnostic criterion for disorders such as those described above?	39.4
24	Should people with disorders such as those described above be clearly informed by their doctors about their diagnosis?	87.9
25	Should relatives of those people be clearly informed by the doctors about the patient's diagnosis?	23.9
35	What does "binge-eating" mean?	85.2
36	What does "compensatory behavior" mean?	94.6
37	What does "restrictive behavior" mean?	98.5
38	May achieving a normal body weight be enough to recover from disorders such as those described above?	70.7

### Table 2 (continued)

ITEM		Frequen- cy of right answers (%)
53	At what age disorders such as those described above most frequently occur?	99
Aetiopo	athogenesis	
30	Is there any evidence that the behavior of family members is always the cause of disorders such as those described above?	32.5
34	Compared to the general population, is an individual with a family member suffering from a disorder such as those described above more likely to develop the disorder?	67
43	Do disorders such as those described above affect only women?	90.9
44	Does excessive vanity contribute to the onset of disorders such as those described above?	89.2
45	Does low self-esteem contribute to the onset of disorders such as those described above?	45.1
46	Does pathological perfectionism contribute to the onset of disorders such as those described above?	45.8
47	Does excessive sensitivity to external judgment contribute to the onset of disorders such as those described above?	53.9
48	Does drug abuse contribute to the onset of disorders such as those described above?	62.3
49	Does friendship with wrong or untrustworthy people contribute to the onset of disorders such as those described above?	60.1
50	Do beauty models promoted by the mass media contribute to the onset of disorders such as those described above?	88.9
51	Does dieting contribute to the onset of disorders such as those described above?	78.3
52	Do stressful life events (breaking up with a partner, overwork, unemployment, birth of a child) contribute to the onset of disor- ders such as those described above?	97.3

 Table 3
 Comparison of error rates among health professional areas

Item	Chi Valu	ie p	Post-hoc test with Bonferroni correction			
9	27.815	0.000	Surg, Med, G	ien > Psyl		
			Surg > Psyt			
10	19.511	0.002	Med > Psyl			
11	14.453	0.013	Surg, Med, G	Gen > Psyl		
17	19.088	0.002	Psyl > All are	as		
18	3.273	0.658				
22	44.469	0.000	All areas > Ps	syt		
25	2.633	0.756				
26	7.667	0.176				
28	16.283	0.006	Surg > Med			
30	9.695	0.084				
32	3.954	0.556				
34	14.347	0.014	Med > Psyt			
39	6.068	0.300				
40	6.821	0.234				
42	3.725	0.542				
45	4.787	0.442				
46	12.089	0.034				
47	3.121	0.681				
48	21.849	0.001	Surg, Gen > I	Psyl		
49	25.983	0.000	Gen > Psyt, F	Psyl		
Gen=0	general	practitioner	area; N	/led=internal	medicine	area;

Psyl=psychological area; Psyt=psychiatric area; Surg=surgical area

associated with a delay in access to specialized care [11]. The most frequent mistake among health professionals compared to psychiatrists is the inclusion of amenorrhea among the diagnostic criteria for AN, which may reflect the DSM-5 nosological change [22] and the need for continuing updates in medical education. It is remarkable that all health professionals, including psychiatrists,

believe that blood tests can help rule out an ED diagnosis, outlining the excessive importance given to biological markers for diagnosis in psychiatry. In addition, another frequent mistake in the entire sample is that relatives of individuals with EDs should not be clearly informed by the doctors about the patient's disorders: this is an important educational gap in light of the contribution that caregivers and family members give to ED diagnosis, access to specialist ED units and recovery [11, 26, 27].

The difficulties with diagnosis are similar to those that occurred when learning about risk factors and the psychopathology of EDs: this is the area with the highest number of items with correct answers reported by less than two thirds of the participants, regardless of their age. Low self-esteem, maladaptive perfectionism, and sensitivity to others' judgments are not considered among the contributing factors to ED vulnerability, while eating behaviors of family members, having a family member suffering from an ED, drug abuse and having friendship with wrong or untrustworthy people are considered among the causes of EDs. On the other hand, respondents correctly acknowledged ED symptoms (e.g., binge-purging behaviors). This outlines an inadequate knowledge of ED etiopathogenesis [28, 29] and psychopathology that is at odds with the recent trend towards a reconceptualization of EDs as affective disorders with high importance of interpersonal problems [30-34]. Educational training should not only focus on the specific eating symptoms, but also on the processes that explain and promote ED symptoms [35]. This is true for all health professionals, including psychiatrists, who are more confident in certain areas (such as the causal role of family

members, drug abuse and friendships) but not in all of them.

Treatment knowledge is another important area investigated in this study. Although psychiatrists are more confident in this area than other health professionals, they report the most frequent mistakes that occurred in the other groups: lack of effectiveness of psychotherapy, lack of knowledge of psychotherapeutic interventions specifically validated for EDs, consideration of the nutritional intervention as the first step in all individuals with EDs, and lack of the need to include family members in the treatment. These findings are in line with those suggesting that psychiatrists lack sufficient knowledge of both validated non-CBT psychotherapies for EDs and use of SSRIs in AN [19]. Evidence-based psychotherapies play a central role in the multidisciplinary approach to EDs [1]; thus, inadequate knowledge of psychotherapy interventions is a gap that needs to be addressed by educational programs. This aligns EDs to other psychiatric disorders where psychotherapy was found to be as effective as pharmacotherapy [36], supporting the need to disseminate psychotherapy effectiveness through educational training.

Treatment of EDs also includes handling physical complications: all health professionals performed well in this area, reporting errors for each item below 33%. It is noteworthy that all participants performed worse than psychiatrists, as non-mental health professionals should have a major role in diagnosing and treating physical complications.

Remarkably, surgeons were more stigmatized towards EDs than psychiatrists. The items that exhibited the most stigma were the following: assuming that individuals with EDs have chosen their abnormal eating behaviors; considering EDs not like any other illness; and making current and lifetime diagnoses based on the external physical appearance of the individual. However, no difference emerged for these items among the study groups. These findings should be considered in light of the observed impaired knowledge of EDs among clinicians and in the context of literature. Previous findings indicate that clinicians often perceive individuals with EDs responsible for causing their illness and prefer dealing with individuals with schizophrenia rather than those with EDs [37]. This is consistent with the more sympathetic attitudes that are observed towards illnesses that are believed to be 'biologically' caused, such as schizophrenia [37]. Overall, the present findings suggest that here may be a stigma against EDs that comes from health professionals and needs to be addressed in addition to the more well-known social stigma [38, 39].

Two general considerations are worth noting: first, better diagnosis and treatment knowledge was associated with older participants, suggesting that clinical experience may play a significant role in addition to educational training. Second, health professionals' perception of family members is highly surprising: the importance of informing them about their relative's diagnosis and including them in the treatment was often neglected, whereas they were often considered to a risk factor for EDs. This picture is in contrast with the increasing evidence that family-based treatments are the most effective in adolescents and young adults with EDs [40]. Educational programs should aim to inform clinicians that aiding caregivers in improving their responses to illness is crucial for recovery [41, 42].

Limitations of the study need to be acknowledged. First, the recruitment procedure, which included health professionals working in a unique university hospital, limits the generalizability of the study findings. Future studies are needed to determine if these findings are replicated in larger samples, including participants who work in different settings. Second, participants' experience in the ED field was not evaluated, which could have had an impact on the study findings. Third, the reliability of subscores was not adequate: although we have partially faced this issue exploring the single item differences among the study groups, the validity of the hypothesized constructs of the questionnaire was not reliable in the present sample. Fourth, the identification of items in the questionnaire may be questionable: some items (namely, those referring to the management of physical complications) could refer to skills that exceed the required education for psychologists and do not represent a clinically significant gap, while others covering important area in the multidisciplinary approach to EDs (namely, the knowledge of severity criteria to define the level or the setting of care) have not been included. Likewise, the evaluation of diagnostic ability did not consider knowledge of diagnostic criteria: the primary focus was on the ability to identify ED signs and symptoms and their management as a general skill. Thus, conclusions about the ability to make early diagnosis must be interpreted with this limitation in mind.

# Conclusions

This is the first study to assess specific gaps in knowledge about the diagnosis, treatment and psychopathology of EDs among health professionals with different specializations. Compared to psychiatrists, other health professionals showed a certain degree of stigma towards EDs: these patients were seen as different from others and held responsible for their abnormal eating behaviors. In addition, inadequate knowledge of general psychopathology and risk factors associated with EDs, insufficient awareness of evidence-based psychotherapy for EDs and neglecting the central role that families play in promoting

# diagnosis and recovery are major gaps that needs to be addressed through education programs.

### Acknowledgements

Not applicable.

### Author contributions

Conceptualization, AMM, AF; Data curation, AP, EB, MC; Formal analysis, GC, MC; Investigation, CB, LL; Methodology, GC, GS, EB; Supervision, AMM, AF; Writing – original draft, AMM; Writing – review & editing, AMM, AF. All the authors have read and approved the final manuscript.

### Funding

This study was supported by the Italian Ministry of Health—Young Researcher Grant (GR-2018-12368336).

### Data availability

The data that support the findings of this study are available from the corresponding author upon reasonable request.

# Declarations

### **Ethics approval**

Ethics approval was not required for this kind of survey as per local legislation and national guidelines.

### **Consent for publication**

Not applicable.

### Competing interests

The authors declare no competing interests.

### Author details

<sup>1</sup>Department of Psychiatry, University of Campania "Luigi Vanvitelli", Largo Nadonna delle Grazie, Naples 80138, Italy

<sup>2</sup>Department of Medicine, Surgery and Dentistry 'Scuola Medica Salernitana', Section of Neurosciences, University of Salerno, Salerno, Italy

### Received: 15 September 2023 / Accepted: 19 June 2024 Published online: 28 June 2024

### References

- Treasure J, Hübel C, Himmerich H. The evolving epidemiology and differential etiopathogenesis of eating disorders: implications for prevention and treatment. World Psychiatry [Internet]. 2022 Feb 1 [cited 2022 Dec 13];21(1):147–8. https://onlinelibrary.wiley.com/doi/full/https://doi.org/10.1002/wps.20935.
- Treasure J, Duarte TA, Schmidt U. Eating disorders. Vol. 395, the Lancet. Lancet Publishing Group; 2020. pp. 899–911.
- Solmi M, Monaco F, Højlund M, Monteleone AM, Trott M, Firth J, et al. Outcomes in people with eating disorders: a transdiagnostic and disorderspecific systematic review, meta-analysis and multivariable meta-regression analysis. World Psychiatry. 2024;23(1):124–38.
- Chesney E, Goodwin GM, Fazel S. Risks of all-cause and suicide mortality in mental disorders: a meta-review. World Psychiatry [Internet]. 2014 [cited 2023 Aug 24];13(2):153.
- Vall E, Wade TD. Predictors of treatment outcome in individuals with eating disorders: A systematic review and meta-analysis [Internet]. Vol. 48, International Journal of Eating Disorders. John Wiley and Sons Inc.; 2015 [cited 2020 Apr 4]. pp. 946–71. http://www.ncbi.nlm.nih.gov/pubmed/26171853.
- Treasure J, Willmott D, Ambwani S, Cardi V, Clark Bryan D, Rowlands K et al. Cognitive Interpersonal Model for Anorexia Nervosa Revisited: The Perpetuating Factors that Contribute to the Development of the Severe and Enduring Illness. J Clin Med [Internet]. 2020 Feb 27 [cited 2020 Mar 20];9(3):630. https:// www.mdpi.com/2077-0383/9/3/630.
- Hilbert A, Hoek H, Schmidt R. Evidence-based clinical guidelines for eating disorders: international comparison. Curr Opin Psychiatry [Internet]. 2017 Nov 1 [cited 2021 Jul 19];30(6):423–37. https://pubmed.ncbi.nlm.nih. gov/28777107/.

- Ali K, Farrer L, Fassnacht DB, Gulliver A, Bauer S, Griffiths KM. Perceived barriers and facilitators towards help-seeking for eating disorders: a systematic review. Int J Eat Disord. 2017;50(1):9–21.
- Monteleone AM, Fernandez-Aranda F, Voderholzer U. Evidence and perspectives in eating disorders: a paradigm for a multidisciplinary approach. World Psychiatry [Internet]. 2019 Oct [cited 2021 Nov 15];18(3):369.
- McGorry PD, Mei C, Chanen A, Hodges C, Alvarez-Jimenez M, Killackey E, Designing. and scaling up integrated youth mental health care. World Psychiatry [Internet]. 2022 Feb 1 [cited 2023 Jan 31];21(1):61–76. https:// pubmed.ncbi.nlm.nih.gov/35015367/.
- Monteleone AM, Barone E, Cascino G, Schmidt U, Gorwood P, Volpe U et al. Pathways to eating disorder care: A European multicenter study. Eur Psychiatry [Internet]. 2023 [cited 2023 Aug 24];66(1). https://pubmed.ncbi.nlm.nih. gov/37092677/.
- Ogg EC, Millar HR, Pusztai EE, Thom AS. General practice consultation patterns preceding diagnosis of eating disorders. Int J Eat Disord. 1997;22(1):89–93.
- Patton GC, Coffey C, Carlin JB, Sanci L, Sawyer S. Prognosis of adolescent partial syndromes of eating disorder. Br J Psychiatry. 2008;192(4):294–9.
- Schmidt U, Ali S, Slone G, Tiller J, Treasure J. The eating disorders awareness test: a new instrument for the assessment of the effectiveness of psychoeducational approaches to the treatment of eating disorders. Eur Eat Disorders Rev. 1995;3(2):103–10.
- Linville D, Benton A, O'Neil M, Sturm K. Medical providers' screening, training and intervention practices for eating disorders. Eat Disord [Internet]. 2010 Mar [cited 2023 Aug 24];18(2):110–31. https://pubmed.ncbi.nlm.nih. gov/20390615/.
- Thompson-Brenner H, Satir DA, Franko DL, Herzog DB. Clinician reactions to patients with eating disorders: a review of the literature. Psychiatr Serv [Internet]. 2012 Jan 1 [cited 2023 Aug 24];63(1):73–8. https://pubmed.ncbi.nlm.nih. gov/22227763/.
- Tse A, Xavier S, Trollope-Kumar K, Agarwal G, Lokker C. Challenges in eating disorder diagnosis and management among family physicians and trainees: a qualitative study. J Eat Disord [Internet]. 2021;10:45. https://doi.org/10.1186/ s40337-022-00570-5.
- Ayton A, Ibrahim A. Does UK medical education provide doctors with sufficient skills and knowledge to manage patients with eating disorders safely? Postgrad Med J [Internet]. 2018 Jun 9 [cited 2023 Aug 24];94(1113):374–80. https://pubmed.ncbi.nlm.nih.gov/29866707/.
- Jones WR, Saeidi S, Morgan JF. Knowledge and attitudes of psychiatrists towards eating disorders. Eur Eat Disord Rev [Internet]. 2013 Jan [cited 2023 Aug 24];21(1):84–8. https://pubmed.ncbi.nlm.nih.gov/23350077/.
- Magliano L, De Rosa C, Fiorillo A, Malangone C, Guarner M, Marasco C et al. [Psychosocial causes and consequences of schizophrenia: opinions of Italians]. Epidemiol Psichiatr Soc [Internet]. 2003 [cited 2023 Sep 10];12(3):187– 97. https://pubmed.ncbi.nlm.nih.gov/14610854/.
- Quaderni del Ministero della Salute. Appropriatezza clinica, strutturale e operativa nella prevenzione, diagnosi e terapia dei disturbi dell'alimentazione.
   2013 [cited 2023 Sep 14]; https://www.salute.gov.it/imgs/C\_17\_pubblicazioni\_2561\_allegato.pdf.
- First MB, Gaebel W, Maj M, Stein DJ, Kogan CS, Saunders JB, et al. An organization-and category-level comparison of diagnostic requirements for mental disorders in ICD-11 and DSM-5. World Psychiatry. 2021;20(1):34–51.
- White M. Sample size in quantitative instrument validation studies: A systematic review of articles published in Scopus, 2021. Heliyon [Internet]. 2022 Dec 1 [cited 2023 Sep 15];8(12). https://pubmed.ncbi.nlm.nih.gov/36568672/.
- 24. Team JASP. JASP. 2023.
- 25. Tu C, Tseng MM, Chang Y. Paths to the first-time diagnoses of anorexia nervosa and bulimia nervosa in Taiwan. Int J Eat Disord. 2021;54(1):59–68.
- Treasure J, Parker S, Oyeleye O, Harrison A. The value of including families in the treatment of anorexia nervosa. Eur Eat Disorders Rev. 2021;29(3):393–401.
- Anastasiadou D, Medina-Pradas C, Sepulveda AR, Treasure J. A systematic review of family caregiving in eating disorders. Eat Behav [Internet]. 2014 [cited 2021 Jan 29];15(3):464–77. https://pubmed.ncbi.nlm.nih. gov/25064301/.
- Morris AS, Hays-Grudo J. Protective and compensatory childhood experiences and their impact on adult mental health. World Psychiatry [Internet]. 2023 Feb 1 [cited 2024 May 30];22(1):150.
- Treasure J, Hübel C, Himmerich H. The evolving epidemiology and differential etiopathogenesis of eating disorders: implications for prevention and treatment. World Psychiatry [Internet]. 2022;21(1):147–8. https://onlinelibrary. wiley.com/doi/full/10.1002/wps.20935.

- Monteleone AM, Cascino G. A systematic review of network analysis studies in eating disorders: is time to broaden the core psychopathology to non specific symptoms. European Eating Disorders Review. Volume 29. John Wiley and Sons Ltd; 2021. pp. 531–47.
- MONTELEONE AM, MARCIELLO F, CASCINO G, ABBATE-DAGA G, ANSELMETTI S, BAIANO M, et al. The impact of COVID-19 lockdown and of the following re-opening period on specific and general psychopathology in people with eating disorders: the emergent role of internalizing symptoms. J Affect Disord. 2021;285:77–83.
- 32. Monteleone AM, Cascino G, Barone E, Carfagno M, Monteleone P. COVID-19 pandemic and eating disorders: what can we learn about psychopathology and treatment? A systematic review. Curr Psychiatry Rep. 2021;23(12):83.
- Gratz KL, Tull MT. A clinically useful conceptualization of emotion regulation grounded in functional contextualism and evolutionary theory. World Psychiatry. 2022;21(3):460–1.
- Watson D, Levin-Aspenson HF, Waszczuk MA, Conway CC, Dalgleish T, Dretsch MN, et al. Validity and utility of hierarchical taxonomy of psychopathology (HiTOP): III. Emotional dysfunction superspectrum. World Psychiatry. 2022;21(1):26–54.
- Oldershaw A, Startup H, Lavender T. Anorexia nervosa and a lost emotional self: a psychological formulation of the development, maintenance, and treatment of Anorexia Nervosa. Front Psychol. 2019;10:1–22.
- 36. Leichsenring F, Steinert C, Rabung S, Ioannidis JPA. The efficacy of psychotherapies and pharmacotherapies for mental disorders in adults: an umbrella review and meta-analytic evaluation of recent meta-analyses. World Psychiatry [Internet]. 2022 Feb 1 [cited 2023 Jan 31];21(1):133–45. https:// onlinelibrary.wiley.com/doi/full/https://doi.org/10.1002/wps.20941.
- 37. Fleming J, Szmukler Gl. Attitudes of medical professionals towards patients with eating disorders. Australasian Psychiatry. 1992;26(3):436–43.

- Pearl RL, Austin SB, Jebeile H, D'Adamo L, Wilfley DE. Opportunities to advance research, intervention, and policy on stigma, eating disorders, and body image. J Eat Disord [Internet]. 2023 Dec 1 [cited 2023 Aug 24];11(1). https://pubmed.ncbi.nlm.nih.gov/37553695/.
- Brelet L, Flaudias V, Désert M, Guillaume S, Llorca PM, Boirie Y. Stigmatization toward People with Anorexia Nervosa, Bulimia Nervosa, and Binge Eating Disorder: A Scoping Review. Nutrients [Internet]. 2021 Aug 1 [cited 2023 Aug 24];13(8).
- Monteleone AM, Pellegrino F, Croatto G, Carfagno M, Hilbert A, Treasure J et al. Treatment of eating disorders: A systematic meta-review of meta-analyses and network meta-analyses. Neurosci Biobehav Rev [Internet]. 2022 Nov 1 [cited 2022 Dec 15];142. https://pubmed.ncbi.nlm.nih.gov/36084848/.
- Treasure J, Nazar BP. Interventions for the Carers of Patients With Eating Disorders. Curr Psychiatry Rep [Internet]. 2016;18(2):16. http://www.ncbi.nlm. nih.gov/pubmed/26781554.
- 42. Treasure J, Schmidt U. The cognitive-interpersonal maintenance model of anorexia nervosa revisited: a summary of the evidence for cognitive, socio-emotional and interpersonal predisposing and perpetuating factors. J Eat Disord [Internet]. 2013;1(1):13. http://jeatdisord.biomedcentral.com/ articles/https://doi.org/10.1186/2050-2974-1-13.

# **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.